Author's response to reviews

Title: Rural-to-urban migration and its implication for New Cooperative Medical Scheme coverage and utilization in China

Authors:

Peiyuan Qiu (qiupeiyuan0218@gmail.com)
Yang Yang (yangyang@scu.edu.cn)
Juying Zhang (juying109@163.com)
Xiao Ma (huaxihe2009@gmail.com)

Version: 8 Date: 29 June 2011

Author's response to reviews:

1. We added "who had the NCMS in 2006" in Line8-9 of the Result section in the Abstract to make it clearer.
2. We added "who had the NCMS in 2006" in Line4 of the first paragraph in 3.3 section.
3. "are" in Line 6 of the first graph of 2.4 section was deleted. (Grammer error)
4. We added a note ("a") for the Table 6 to make it clearer.
5. "0=not enrollment and 1=enrollment" in the Table 5 was changed to "0=non-enrollment and 1=enrollment"